Author's response to reviews

Title: The Survey of the Health of Wisconsin (SHOW), a novel infrastructure for population health research: rationale and methods

Authors:

F. Javier Nieto (fjnieto@wisc.edu)
Paul E Peppard (ppeppard@wisc.edu)
Corinne D Engelman (cengelman@wisc.edu)
Jane A McElroy (mcelroyja@health.missouri.edu)
Loren W Galvao (lgalvao@uwm.edu)
Elliot M Friedman (friedman1@wisc.edu)
Andrew J Bersch (abersch@show.wisc.edu)
Kristen C Malecki (kmalecki@wisc.edu)

Version: 5 Date: 8 December 2010

Author's response to reviews:

In the revised version, typographical and slight grammar errors have been corrected. We have updated the number of participants recruited in 2010 (now 940), edited the acknowledgment sections and the legend for figure 2. We've replaced figure 2 for a higher resolution picture.

AS PER EMAIL SENT A FEW MINUTES AGO, HOWEVER, WE'D APPRECIATE IT IF YOU COULD GIVE US AN ADDITIONAL 24-48 HOUR EXTENSION TO POST THE FINAL VERSION OF THE MANUSCRIPT ONCE WE REVISED THE NUMBERS IN TABLE 2, AS AN ERROR IN THE APPLIED SAMPLING WEIGHTS DURING THE ANALYSES, MIGHT SLIGHTLY CHANGE A FEW OF THE NUMBERS.

THANKS.